Author’s response to reviews

Title: Women born with very low birth weight have similar menstrual cycle pattern, pregnancy rates and hormone profiles compared with women born at term

Authors:

Gunilla Sydsjö (Gunilla.Sydsjo@liu.se)
Pia Törnblom (Pia.Tornblom@liu.se)
P-O Gäddlin (po.gaddlin@telia.com)
Orvar Finnström (Orvar.Finnsstrom@liu.se)
Ingemar Leijon (Ingeamr.Leijon@liu.se)
Nina Nelson (Nina.Nelson@liu.se)
Elvar Theodorsson (Elvar.Theodorsson@liu.se)
Mats Hammar (Mats.Hammar@liu.se)

Version: 2 Date: 08 Apr 2019

Author’s response to reviews:

Thank you for this nice reply.

I cannot provide a translated version of the questioner - all the questions and answers is provided in the tables so the readers are able to see this.

You will have a clean copy with the changes in a day.

Thank you for the handling of this manuscript., Efficient and polite.

Kind regards Gunilla Sydsjö